Author's response to reviews

Title: Sick-listed employees with severe medically unexplained physical symptoms: burden or routine for the occupational health physician? A cross-sectional study.

Authors:

Rob Hoedeman (rob.hoedeman@arboned.nl)
Boudien Krol (b.krol@med.umcg.nl)
Annette H Blankenstein (ah.blankenstein@vumc.nl)
Petra C Koopmans (petra.koopmans@arboned.nl)
Johan W Groothoff (j.w.groothoff@med.umcg.nl)

Version: 4 Date: 1 November 2010

Author's response to reviews: see over
Dear Editors,

We performed the requested changes. Please find below our specific responses to the comments.

Major revisions:

We adjusted the list and names of authors, as registered in the submission system.

Minor revisions:

- We removed all highlighting and tracking.
- The section ‘Subject and methods’ was renamed as ‘Methods’
- Tables: we removed all vertical lines.
- Typography: we checked all content and found 2 typos, which we corrected.

We hope that you will find our corrections satisfactory, and look forward to your reply.

Yours sincerely,
Rob Hoedeman
Corresponding author